Author's response to reviews

Title: Associations between Home and Ambulatory Blood Pressure Measurements and Cardiovascular Risk

Authors:

Sigrun A Chrubasik (sigrun.chrubasik@klinikum.uni-freiburg.de)
Cosima A Chrubasik (sigrun.chrubasik@resortragaz.ch)
Brigitte Walz (Brigitte.Walz@ksgr.ch)
Juergen Schulte Mönning (jsm@imbi.uni-freiburg.de)
Paul Erne (paul.erne@erne-net.ch)

Version: 2 Date: 30 January 2014

Author's response to reviews:

We have done as requested:
made transparent the competing interests
specified the contributions of the authors
included the legends and the Tables into the main document